Author's response to reviews

Title: Complete Resection of Undifferentiated Cardiac Sarcoma and Reconstruction of the Atria and the Superior Vena Cava: Case Report

Authors:

Nobuyuki Furukawa (gtgkk994@yahoo.co.jp)
Jan Gummert (jgummert@hdz-nrw.de)
Jochen Börgermann (jboergermann@hdz-nrw.de)

Version: 2 Date: 21 July 2012

Author's response to reviews: see over
Dear Dr. Vipin Zamvar:

The comments of the two reviewers have been helpful in allowing us to revise our manuscript. We have attempted to address the questions raised by the referees according to the following:

1. We would agree with the reviewer that more technical details is now required regarding the Operation. We have revised the description of the surgery and the perioperative course in detail.
2. The patient is not female but male. We are so sorry for making such a failure.
3. Recently, we had a re-examination of the patient almost 3 years after surgery. The patient is doing well and remains free of metastase or local tumor recurrence. Considering to the poor prognosis of cardiac sarcoma, it is also one of the valuable information. We have rewritten it, “thirty-five months”.
4. We have also improved into more suitable English representations as a whole.

Thank you for your consideration of the revised version.

Sincerely,

Nobuyuki Furukawa, M.D.
Department of Cardiothoracic Surgery, Heart and Diabetes Center North Rhine-Westphalia, Bad Oeynhausen, Germany
Address reprint requests to Dr. Nobuyuki Furukawa, Georgstr. 11, 32545 Bad Oeynhausen, Germany
Department of Cardiothoracic Surgery, Heart and Diabetes Center North Rhine-Westphalia, Bad Oeynhausen, Germany
Phone: +49-(0)5731-97-0, Fax: +49-(0)5731-97-2300
E-mail address: qtgkk994@yahoo.co.jp